Author's response to reviews

Title: Treatment with betablockers is associated with higher gray-scale median in carotid plaques.

Authors:

Giuseppe Asciutto (giuseppe.asciutto@med.lu.se)
Nuno v Dias (nunovdias@gmail.com)
Ana Persson (ana.persson@med.lu.se)
Jan Nilsson (jan.nilsson@med.lu.se)
Isabel Gonçalves (migvdias@gmail.com)

Version: 2
Date: 10 May 2014

Author's response to reviews:

Dear Editors,

the current manuscript has been revised in the following parts:

lines and pages numbers have been included.
lines 106-107: the name of the regional ethical committee as well as the decision´s number have been inserted.
lines 300-306: the section “Author´s contribution” has been added.

Sincerely

The authors